Author's response to reviews

Title: Reconsidering low dose aspirin therapy for cardiovascular disease: a study protocol for physician and patient behavioral change

Authors:

Brittany Folks (Brittany.Folks@ucdenver.edu)
William G. LeBlanc (William.LeBlanc@ucdenver.edu)
Elizabeth W. Staton (Elizabeth.Staton@ucdenver.edu)
Wilson D. Pace (Wilson.Pace@ucdenver.edu)

Version: 6 Date: 13 June 2011

Author's response to reviews:

We have approved the changes made by the copyeditor and added a few similar wording adjustments not related to the content.